Author’s response to reviews

Title: Genotypic determination of resistance and heteroresistance to clarithromycin in Helicobacter pylori isolates from antrum and corpus of Colombian symptomatic patients

Authors:
Betsy Arévalo-Jaimes (bv.arevalo1563@uniandes.edu.co)
Diana Rojas-Rengifo (di-rojas@uniandes.edu.co)
Carlos Jaramillo (cjaramil@uniandes.edu.co)
Belen Mendoza de Molano (bmolano@gmail.com)
Jose Vera-Chamorro (jfvera1@gmail.com)
Maria del Pilar Delgado (mdelgado@uniandes.edu.co)

Version: 2 Date: 01 Apr 2019

Author’s response to reviews:

Bogotá, April 1st 2019

Mrs.
Oliver Cocks
Editor

BMC Infectious Diseases

https://bmcinfectdis.biomedcentral.com/

Ref: Manuscript Submission: INFD-D-18-01256, entitled "Genotypic determination of resistance and heteroresistance to clarithromycin in Helicobacter pylori isolates from antrum and corpus of Colombian symptomatic patients".

I am submitting the final version with minor revision of the above manuscript. The following lists point by point the responses to the Editor and reviewers’ comments:

Technical Comments

I. Editor Comments:
Comment 1. The link to your data in **NCBI** provided in the Availability of Data and Materials statement is currently broken. Please amend this in the revised version of the manuscript.

Answer 1. The link was revised and now it’s functional. Actually, the link allows to be accessed.

Comment 2. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements, please put “Not Applicable” in the Acknowledgements section.

Answer 2. The funding information from the Acknowledgements was moved to the Funding section.

Comment 3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer 3. All revision marks were removed from the manuscript. All materials (manuscript, tables and figures) were revised and are final and clean versions. As you requested, figures remain uploaded as separate files.

II. Reviewer reports

Marja Liisa Hänninen (Reviewer 2). Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format. Please overwrite this text when adding your comments to the authors.

Answer. The authors have revised the manuscript. It was adjusted according to reviewers and editor suggestions. We hope that these responses are satisfactory and look forward to hearing from you. All comments for the authors and the corresponding answers are in this e-mail box as requested.

Yours sincerely,

Maria del Pilar Delgado

Corresponding author